\documentclass{article}
%To use pdflatex, uncomment these lines, as well as the \href lines
%in each entry
%\usepackage[pdftex,
%       colorlinks=true,
%       urlcolor=blue,       % \href{...}{...} external (URL)
%       filecolor=green,     % \href{...} local file
%       linkcolor=red,       % \ref{...} and \pageref{...}
%       pdftitle={Papers by AUTHOR},
%       pdfauthor={Your Name},
%       pdfsubject={Just a test},
%       pdfkeywords={test testing testable},
%       pagebackref,
%       pdfpagemode=None,
%        bookmarksopen=true]{hyperref}
%usepackage{arial}
%\renewcommand{\familydefault}{\sfdefault} % San serif

\pagestyle{empty}
\oddsidemargin 0.0in
\textwidth 6.5in
\topmargin -0.75in
\textheight 9.5in

\begin{document}
\title{Papers by JADE 1979-1991}
\author{}
\date{}
\maketitle
\begin{enumerate}


%%   LIST OF PAPERS
%%   Please comment out anything between here and the
%%   first \item
%%   Please send any updates or corrections to the list to
%%   library@slac.stanford.edu


%\cite{Pust:1991nj}
\item%{Pust:1991nj}
{\bf ``Measurement and analysis of the reaction gamma gamma $\to$ 3 pi+ 3 pi-''}
  \\{}R.~Pust {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 51}, 531 (1991)
%\\{}DESY-91-003
%\href{/spires/find/hep/www?j=zepya\%2cc51\%2c531}{SPIRES entry}

%\cite{Magnussen:1990zg}
\item%{Magnussen:1990zg}
{\bf ``A Comparison Of Measured Jet Cross-Sections With QCD Calculations For E+ E- Annihilation''}
  \\{}N.~Magnussen {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 49}, 29 (1991)
%\\{}DESY-90-089
%\href{/spires/find/hep/www?j=zepya\%2cc49\%2c29}{SPIRES entry}

%\cite{Hagemann:1990dj}
\item%{Hagemann:1990dj}
{\bf ``A Measurement of the average lifetime of B hadrons produced by e+ e- collisions at s**(1/2) = 36.3-GeV''}
  \\{}J.~Hagemann {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 48}, 401 (1990)
%\\{}DESY-90-087
%\href{/spires/find/hep/www?j=zepya\%2cc48\%2c401}{SPIRES entry}

%\cite{Wegner:1990cg}
\item%{Wegner:1990cg}
{\bf ``Measurement of two photon production of rho0 omega''}
  \\{}A.~Wegner {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 48}, 393 (1990)
%\\{}DESY-90-069
%\href{/spires/find/hep/www?j=zepya\%2cc48\%2c393}{SPIRES entry}

%\cite{Oest:1990ki}
\item%{Oest:1990ki}
{\bf ``MEASUREMENT OF RESONANCE PRODUCTIONS IN THE REACTIONS gamma gamma $\to$ pi0 pi0 AND gamma gamma $\to$ pi0 eta''}
  \\{}T.~Oest {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 47}, 343 (1990)
%\\{}DESY 90/025
%\href{/spires/find/hep/www?j=zepya\%2cc47\%2c343}{SPIRES entry}

%\cite{Hegner:1989rd}
\item%{Hegner:1989rd}
{\bf ``FINAL RESULTS ON MU AND TAU PAIR PRODUCTION BY THE JADE COLLABORATION AT PETRA''}
  \\{}S.~Hegner {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 46}, 547 (1990)
%\\{}DESY 89/178
%\href{/spires/find/hep/www?j=zepya\%2cc46\%2c547}{SPIRES entry}

%\cite{Pitzl:1989qy}
\item%{Pitzl:1989qy}
{\bf ``A Study of Photon Production in Hadronic Events from e+ e- Annihilation''}
  \\{}D.~Pitzl {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 46}, 1 (1990)
  [Erratum-ibid.\  C {\bf 47}, 676 (1990)]
%\\{}DESY-89-129
%\href{/spires/find/hep/www?j=zepya\%2cc46\%2c1}{SPIRES entry}

%\cite{Elsen:1989wg}
\item%{Elsen:1989wg}
{\bf ``A Measurement Of The Weak Axial Couplings Of The B And C Quark''}
  \\{}E.~Elsen {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 46}, 349 (1990)
%\\{}DESY-89-127
%\href{/spires/find/hep/www?j=zepya\%2cc46\%2c349}{SPIRES entry}

%\cite{OuldSaada:1989qs}
\item%{OuldSaada:1989qs}
{\bf ``A Measurement Of The Charmed Quark Asymmetry In E+ E- Annihilation''}
  \\{}F.~Ould-Saada {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 44}, 567 (1989)
%\\{}DESY 89/063
%\href{/spires/find/hep/www?j=zepya\%2cc44\%2c567}{SPIRES entry}

%\cite{Greenshaw:1988uv}
\item%{Greenshaw:1988uv}
{\bf ``A MEASUREMENT OF THE CHARGE ASYMMETRY OF HADRONIC EVENTS IN ELECTRON POSITRON ANNIHILATION''}
  \\{}T.~Greenshaw {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 42}, 1 (1989)
%\\{}DESY 88/154
%\href{/spires/find/hep/www?j=zepya\%2cc42\%2c1}{SPIRES entry}

%\cite{Hill:1988my}
\item%{Hill:1988my}
{\bf ``Observation Of Spin 1 Resonance Formation In The Final State K K Pi Produced In Tagged Two Photon Collisions''}
  \\{}P.~Hill {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 42}, 355 (1989)
%\\{}DESY 88/166
%\href{/spires/find/hep/www?j=zepya\%2cc42\%2c355}{SPIRES entry}

%\cite{Kleinwort:1988sb}
\item%{Kleinwort:1988sb}
{\bf ``A MEASUREMENT OF THE tau LEPTON LIFETIME''}
  \\{}C.~Kleinwort {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 42}, 7 (1989)
%\\{}DESY-88-155
%\href{/spires/find/hep/www?j=zepya\%2cc42\%2c7}{SPIRES entry}

%\cite{Bethke:1988zc}
\item%{Bethke:1988zc}
{\bf ``Experimental Investigation of the Energy Dependence of the Strong Coupling Strength''}
  \\{}S.~Bethke {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 213}, 235 (1988)
%\\{}DESY 88/105
%\href{/spires/find/hep/www?j=phlta\%2cb213\%2c235}{SPIRES entry}

%\cite{OuldSaada:1988bc}
\item%{OuldSaada:1988bc}
{\bf ``COMPARISON OF THE PARTICLE FLOW IN q anti-q gamma AND q anti-q g EVENTS FROM e+ e- ANNIHILATIONS AT PETRA''}
  \\{}F.~Ould Saada {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 39}, 1 (1988)
%\\{}DESY-88-015
%\href{/spires/find/hep/www?j=zepya\%2cc39\%2c1}{SPIRES entry}

%\cite{Bartel:1987de}
\item%{Bartel:1987de}
{\bf ``SEARCH FOR LEPTOQUARKS AND OTHER NEW PARTICLES WITH LEPTON - HADRON SIGNATURE IN e+ e- INTERACTIONS''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 36}, 15 (1987)
%\\{}DESY-87-031
%\href{/spires/find/hep/www?j=zepya\%2cc36\%2c15}{SPIRES entry}

%\cite{Bartel:1986mf}
\item%{Bartel:1986mf}
{\bf ``DETERMINATION OF SEMIMUONIC BRANCHING RATIOS AND FRAGMENTATION FUNCTIONS OF HEAVY QUARKS IN e+ e- ANNIHILATION AT s**(1.2) = 34.6-GeV''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 33}, 339 (1987)
%\\{}DESY-86-129
%\href{/spires/find/hep/www?j=zepya\%2cc33\%2c339}{SPIRES entry}

%\cite{Bartel:1986mx}
\item%{Bartel:1986mx}
{\bf ``OBSERVATION OF CHARMED MESONS IN PHOTON-PHOTON COLLISIONS''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 184}, 288 (1987)
%\\{}DESY 86-125
%\href{/spires/find/hep/www?j=phlta\%2cb184\%2c288}{SPIRES entry}

%\cite{Bartel:1986un}
\item%{Bartel:1986un}
{\bf ``MEASUREMENT OF THE BRANCHING FRACTIONS FOR tau $\to$ e electron-neutrino tau-neutrino, tau $\to$ mu muon-neutrino tau-neutrino AND tau
$\to$ pi tau-neutrino''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 182}, 216 (1986)
%\\{}DESY 86/091
%\href{/spires/find/hep/www?j=phlta\%2cb182\%2c216}{SPIRES entry}
\\{}{\it Contributed to Int. Conf. on High Energy Physics, Berkeley, CA, Jul 16-23, 1986}

%\cite{Bartel:1986ua}
\item%{Bartel:1986ua}
{\bf ``Experimental Studies on Multi-Jet Production in e+ e- Annihilation at PETRA Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 33}, 23 (1986)
%\\{}DESY 86/086
%\href{/spires/find/hep/www?j=zepya\%2cc33\%2c23}{SPIRES entry}

%\cite{Bartel:1986sy}
\item%{Bartel:1986sy}
{\bf ``EXCLUSIVE PRODUCTION OF PROTON ANTI-PROTON PAIRS IN PHOTON-PHOTON COLLISIONS''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 174}, 350 (1986)
%\\{}DESY-86-043
%\href{/spires/find/hep/www?j=phlta\%2cb174\%2c350}{SPIRES entry}

%\cite{Bartel:1986ee}
\item%{Bartel:1986ee}
{\bf ``Determination Of The B Lifetime''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 31}, 349 (1986)
%\\{}DESY 86/001
%\href{/spires/find/hep/www?j=zepya\%2cc31\%2c349}{SPIRES entry}

%\cite{Bartel:1986ed}
\item%{Bartel:1986ed}
{\bf ``Lepton Pair Production In Double Tagged Two Photon Interactions''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 30}, 545 (1986)
%\\{}DESY 86/005
%\href{/spires/find/hep/www?j=zepya\%2cc30\%2c545}{SPIRES entry}

%\cite{Bartel:1985cs}
\item%{Bartel:1985cs}
{\bf ``Tests Of The Standard Model In Leptonic Reactions At Petra Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 30}, 371 (1986)
%\\{}DESY-85-131
%\href{/spires/find/hep/www?j=zepya\%2cc30\%2c371}{SPIRES entry}

%\cite{Bartel:1985bf}
\item%{Bartel:1985bf}
{\bf ``A Measurement Of The Mean Semimuonic Branching Ratio Of B Hadrons Produced At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 163}, 277 (1985)
%\\{}DESY-85-071
%\href{/spires/find/hep/www?j=phlta\%2cb163\%2c277}{SPIRES entry}

%\cite{Bartel:1985be}
\item%{Bartel:1985be}
{\bf ``Inclusive Neutral D* Production And Limits On F* Production In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 161}, 197 (1985)
%\\{}DESY-85-069
%\href{/spires/find/hep/www?j=phlta\%2cb161\%2c197}{SPIRES entry}

%\cite{Bartel:1985wq}
\item%{Bartel:1985wq}
{\bf ``Evidence For Iota Production In High-Energy E+ E- Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
\\{}DESY 85/081(1985)
%\href{/spires/find/hep/www?r=desy\%2F85\%2F081}{SPIRES entry}

%\cite{Bartel:1985mh}
\item%{Bartel:1985mh}
{\bf ``Tau Lepton Production And Decay At Petra Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 161}, 188 (1985)
%\\{}DESY-85-065
%\href{/spires/find/hep/www?j=phlta\%2cb161\%2c188}{SPIRES entry}

%\cite{Bartel:1985ey}
\item%{Bartel:1985ey}
{\bf ``A Measurement Of The Total Cross-Section And A Study Of Inclusive Muon Production For The Process E+ E- $\to$ Hadrons In The Energy Range
Between 39.79-Gev And 46.78-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 160}, 337 (1985)
%\\{}DESY 85/057
%\href{/spires/find/hep/www?j=phlta\%2cb160\%2c337}{SPIRES entry}

%\cite{Bartel:1985ex}
\item%{Bartel:1985ex}
{\bf ``A Search For The Supersymmetric Chargino In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 29}, 505 (1985)
%\\{}DESY-85-060
%\href{/spires/find/hep/www?j=zepya\%2cc29\%2c505}{SPIRES entry}

%\cite{Bartel:1985zx}
\item%{Bartel:1985zx}
{\bf ``Comparison Of Three Jet Events With QCD Shower Models''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 157}, 340 (1985)
%\\{}DESY 85/036
%\href{/spires/find/hep/www?j=phlta\%2cb157\%2c340}{SPIRES entry}

%\cite{Bartel:1985zw}
\item%{Bartel:1985zw}
{\bf ``A Measurement Of The Eta Radiative Width Gamma Eta $\to$ Gamma Gamma''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 158}, 511 (1985)
  [Phys.\ Lett.\  B {\bf 160}, 421 (1985)]
%\\{}DESY 85/033
%\href{/spires/find/hep/www?j=phlta\%2cb160\%2c421}{SPIRES entry}

%\cite{Bartel:1985wn}
\item%{Bartel:1985wn}
{\bf ``A Study Of Photon Production In Hadronic E+ E- Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 28}, 343 (1985)
%\\{}DESY 85/029
%\href{/spires/find/hep/www?j=zepya\%2cc28\%2c343}{SPIRES entry}

%\cite{Bartel:1985ik}
\item%{Bartel:1985ik}
{\bf ``A Search For Monojet Events Produced By Virtual Z0 Bosons In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 155}, 288 (1985)
%\\{}DESY 85/022
%\href{/spires/find/hep/www?j=phlta\%2cb155\%2c288}{SPIRES entry}

%\cite{Bartel:1984ag}
\item%{Bartel:1984ag}
{\bf ``Search For Scalar Muons''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 152}, 392 (1985)
%\\{}DESY 84/113
%\href{/spires/find/hep/www?j=phlta\%2cb152\%2c392}{SPIRES entry}

%\cite{Bartel:1984af}
\item%{Bartel:1984af}
{\bf ``Search For Supersymmetric Partners Of Electrons''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 152}, 385 (1985)
%\\{}DESY 84/112
%\href{/spires/find/hep/www?j=phlta\%2cb152\%2c385}{SPIRES entry}

%\cite{Bartel:1984xh}
\item%{Bartel:1984xh}
{\bf ``New Results On E+ E- $\to$ Mu+ Mu- From The Jade Detector At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 26}, 507 (1985)
%\\{}DESY 84/078
%\href{/spires/find/hep/www?j=zepya\%2cc26\%2c507}{SPIRES entry}

%\cite{Bartel:1984rh}
\item%{Bartel:1984rh}
{\bf ``Inclusive Production Of Vector Mesons Rho0 And K*+- In E+ E- Annihilation At  = 35-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 145}, 441 (1984)
%\\{}DESY 84/058
%\href{/spires/find/hep/www?j=phlta\%2cb145\%2c441}{SPIRES entry}

%\cite{Bartel:1984rg}
\item%{Bartel:1984rg}
{\bf ``A Measurement Of The Electroweak Induced Charge Asymmetry In E+ E- $\to$ B Anti-B''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 146}, 437 (1984)
%\\{}DESY-84-068
%\href{/spires/find/hep/www?j=phlta\%2cb146\%2c437}{SPIRES entry}

%\cite{Bartel:1984ud}
\item%{Bartel:1984ud}
{\bf ``Charged D* Production In E+ E- Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 146}, 121 (1984)
%\\{}DESY 84/059
%\href{/spires/find/hep/www?j=phlta\%2cb146\%2c121}{SPIRES entry}

%\cite{Bartel:1984uc}
\item%{Bartel:1984uc}
{\bf ``Measurements Of Energy Correlations In E+ E- $\to$ Hadrons''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 25}, 231 (1984)
%\\{}DESY 84/050
%\href{/spires/find/hep/www?j=zepya\%2cc25\%2c231}{SPIRES entry}

%\cite{Bartel:1984hv}
\item%{Bartel:1984hv}
{\bf ``Observation Of Radiative E+ E- Annihilation Into Mu Mu Gamma And Mu Mu Gamma Gamma''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 24}, 223 (1984)
%\\{}DESY 84/033
%\href{/spires/find/hep/www?j=zepya\%2cc24\%2c223}{SPIRES entry}

%\cite{Bartel:1984cg}
\item%{Bartel:1984cg}
{\bf ``Experimental Study Of The Photon Structure Function F(2) At Q**2 From 10-Gev**2 To 220-Gev**2''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 24}, 231 (1984)
%\\{}DESY 84/042
%\href{/spires/find/hep/www?j=zepya\%2cc24\%2c231}{SPIRES entry}

%\cite{Bartel:1984cf}
\item%{Bartel:1984cf}
{\bf ``A Search For The Supersymmetric Partner Of The Z0 In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 146}, 126 (1984)
%\\{}DESY 84/038
%\href{/spires/find/hep/www?j=phlta\%2cb146\%2c126}{SPIRES entry}

%\cite{Bartel:1984ix}
\item%{Bartel:1984ix}
{\bf ``A Search For Massive Photinos At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 139}, 327 (1984)
%\\{}DESY 84/016
%\href{/spires/find/hep/www?j=phlta\%2cb139\%2c327}{SPIRES entry}

%\cite{Bartel:1983iq}
\item%{Bartel:1983iq}
{\bf ``A Search For Flavor Changing Neutral Currents In B Decay At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 132}, 241 (1983)
%\\{}DESY 83/049
%\href{/spires/find/hep/www?j=phlta\%2cb132\%2c241}{SPIRES entry}

%\cite{Bartel:1983ij}
\item%{Bartel:1983ij}
{\bf ``Test Of Fragmentation Models By Comparison With Three Jet Events Produced In E+ E- $\to$ Hadrons''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 134}, 275 (1984)
%\\{}DESY 83/079
%\href{/spires/find/hep/www?j=phlta\%2cb134\%2c275}{SPIRES entry}

%\cite{Bartel:1983ii}
\item%{Bartel:1983ii}
{\bf ``Particle Distribution In Three Jet Events Produced By E+ E- Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 21}, 37 (1983)
%\\{}DESY-83-080
%\href{/spires/find/hep/www?j=zepya\%2cc21\%2c37}{SPIRES entry}

%\cite{Bartel:1983pe}
\item%{Bartel:1983pe}
{\bf ``Observation Of Inclusive Eta Production In E+ E- Annihilation At Cm Energies Of 34-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 130}, 454 (1983)
%\\{}DESY-83-063
%\href{/spires/find/hep/www?j=phlta\%2cb130\%2c454}{SPIRES entry}

%\cite{Bartel:1983cv}
\item%{Bartel:1983cv}
{\bf ``Measurement Of The Processes E+ E- $\to$ E+ E- And E+ E- $\to$ Gamma Gamma At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 19}, 197 (1983)
%\\{}DESY 83/035
%\href{/spires/find/hep/www?j=zepya\%2cc19\%2c197}{SPIRES entry}

%\cite{Bartel:1983ym}
\item%{Bartel:1983ym}
{\bf ``Precise Measurement Of Total Cross-Sections For The Process E+ E- $\to$ Multi-Hadrons In The Center-Of-Mass Energy Range Between 12.0-Gev
And 36.4-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 129}, 145 (1983)
%\\{}DESY-83-050
%\href{/spires/find/hep/www?j=phlta\%2cb129\%2c145}{SPIRES entry}

%\cite{Bartel:1983qp}
\item%{Bartel:1983qp}
{\bf ``Charged Particle And Neutral Kaon Production In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 20}, 187 (1983)
%\\{}DESY 83/042
%\href{/spires/find/hep/www?j=zepya\%2cc20\%2c187}{SPIRES entry}

%\cite{Bartel:1983ud}
\item%{Bartel:1983ud}
{\bf ``A Search For Heavy Leptons At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 123}, 353 (1983)
%\\{}DESY 83/002
%\href{/spires/find/hep/www?j=phlta\%2cb123\%2c353}{SPIRES entry}

%\cite{Bartel:1982ub}
\item%{Bartel:1982ub}
{\bf ``Experimental Evidence For Differences In P(T) Between Quark Jets And Gluon Jets''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 123}, 460 (1983)
%\\{}DESY 82/086
%\href{/spires/find/hep/www?j=phlta\%2cb123\%2c460}{SPIRES entry}

%\cite{Bartel:1982ix}
\item%{Bartel:1982ix}
{\bf ``Experimental Study Of The Photon Structure Function F(2) In The High Q**2 Region''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 121}, 203 (1983)
%\\{}DESY 82/064
%\href{/spires/find/hep/www?j=phlta\%2cb121\%2c203}{SPIRES entry}

%\cite{Bartel:1982qm}
\item%{Bartel:1982qm}
{\bf ``Differential Three Jet Cross-Section In E+ E- Annihilation And Comparison With Second Order Predictions Of QCD And Abelian Vector
Theory''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 119}, 239 (1982)
%\\{}DESY 82/060
%\href{/spires/find/hep/www?j=phlta\%2cb119\%2c239}{SPIRES entry}

%\cite{Bartel:1982aw}
\item%{Bartel:1982aw}
{\bf ``A Search For Charged Scalar Particles Pair Produced In E+ E- Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 114}, 211 (1982)
%\\{}DESY 82/023
%\href{/spires/find/hep/www?j=phlta\%2cb114\%2c211}{SPIRES entry}

%\cite{Bartel:1982fs}
\item%{Bartel:1982fs}
{\bf ``Upper Limit On Beauty Lifetime And Lower Limit On Weak Mixing Angles''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 114}, 71 (1982)
%\\{}DESY 82/014
%\href{/spires/find/hep/www?j=phlta\%2cb114\%2c71}{SPIRES entry}

%\cite{Bartel:1982fk}
\item%{Bartel:1982fk}
{\bf ``Observation Of Four - Jet Structure In E+ E- Annihilation At S**(1/2) = 33-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 115}, 338 (1982)
%\\{}DESY-82-016
%\href{/spires/find/hep/www?j=phlta\%2cb115\%2c338}{SPIRES entry}

%\cite{Bartel:1982qd}
\item%{Bartel:1982qd}
{\bf ``A Measurement Of The Reaction E+ E- $\to$ E+ E- Eta-Prime And The Radiative Width Gamma (Eta-Prime $\to$ Gamma Gamma) At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 113}, 190 (1982)
%\\{}DESY 82/007
%\href{/spires/find/hep/www?j=phlta\%2cb113\%2c190}{SPIRES entry}

%\cite{Bartel:1981gz}
\item%{Bartel:1981gz}
{\bf ``Observation Of A Charge Asymmetry In E+ E- $\to$ Mu+ Mu-''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 108}, 140 (1982)
%\\{}DESY 81/072
%\href{/spires/find/hep/www?j=phlta\%2cb108\%2c140}{SPIRES entry}

%\cite{Bartel:1981kt}
\item%{Bartel:1981kt}
{\bf ``Observation Of High P(T) Jets In Two Photon Interactions''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 107}, 163 (1981)
%\\{}DESY 81/048
%\href{/spires/find/hep/www?j=phlta\%2cb107\%2c163}{SPIRES entry}

%\cite{Bartel:1981sw}
\item%{Bartel:1981sw}
{\bf ``Baryon Production In E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 104}, 325 (1981)
%\\{}DESY 81/028
%\href{/spires/find/hep/www?j=phlta\%2cb104\%2c325}{SPIRES entry}

%\cite{Bartel:1981dk}
\item%{Bartel:1981dk}
{\bf ``Energy Carried By Gamma Rays And Neutral Particles In Multi - Hadron Final States At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 9}, 315 (1981)
%\\{}DESY 81/025
%\href{/spires/find/hep/www?j=zepya\%2cc9\%2c315}{SPIRES entry}

%\cite{Bartel:1981xt}
\item%{Bartel:1981xt}
{\bf ``A Determination Of Quark Weak Couplings At Petra Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 101}, 361 (1981)
%\\{}DESY 81/015
%\href{/spires/find/hep/www?j=phlta\%2cb101\%2c361}{SPIRES entry}

%\cite{Bartel:1981kh}
\item%{Bartel:1981kh}
{\bf ``Experimental Study Of Jets In Electron - Positron Annihilation''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 101}, 129 (1981)
%\\{}DESY-81-009
%\href{/spires/find/hep/www?j=phlta\%2cb101\%2c129}{SPIRES entry}

%\cite{Bartel:1981gza}
\item%{Bartel:1981gza}
{\bf ``Search For Narrow Resonances In E+ E- Annihilation At C.M. Energies Between 33.0-Gev And 36.72-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 100}, 364 (1981)
%\\{}DESY-81-006
%\href{/spires/find/hep/www?j=phlta\%2cb100\%2c364}{SPIRES entry}

%\cite{Bartel:1980qc}
\item%{Bartel:1980qc}
{\bf ``Experimental Limits On The Strength Of Weak Neutral Currents In Lepton Pair Production At Petra Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 99}, 281 (1981)
%\\{}DESY 80/123
%\href{/spires/find/hep/www?j=phlta\%2cb99\%2c281}{SPIRES entry}

%\cite{Bartel:1980hg}
\item%{Bartel:1980hg}
{\bf ``Prompt Muons In Multiparticle Events From E+ E- Annihilation At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 99}, 277 (1981)
%\\{}DESY 80/86
%\href{/spires/find/hep/www?j=phlta\%2cb99\%2c277}{SPIRES entry}

%\cite{Bartel:1980zw}
\item%{Bartel:1980zw}
{\bf ``Search For Fractional Charge And Heavy Stable Particles At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Z.\ Phys.\  C {\bf 6}, 295 (1980)
%\\{}DESY-80-71
%\href{/spires/find/hep/www?j=zepya\%2cc6\%2c295}{SPIRES entry}

%\cite{Bartel:1980vf}
\item%{Bartel:1980vf}
{\bf ``Results From Jade''}
  \\{}W.~Bartel and A.~Petersen  [JADE Collaboration]
\\{}DESY-80-46(1980)
%\href{/spires/find/hep/www?r=desy-80-46}{SPIRES entry}
\\{}{\it Talks given at the 15th Rencontre de Moriond, Les Arcs, France, Mar 9-21, 1980}

%\cite{Bartel:1980it}
\item%{Bartel:1980it}
{\bf ``Test Of Quantum Electrodynamics At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 92}, 206 (1980)
%\\{}DESY 80/14
%\href{/spires/find/hep/www?j=phlta\%2cb92\%2c206}{SPIRES entry}

%\cite{Bartel:1980rr}
\item%{Bartel:1980rr}
{\bf ``Search For Narrow Resonances In E+ E- Annihilation At C.M. Energies Between 29.90-Gev And 31.46-Gev''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 91}, 152 (1980)
%\\{}DESY 80/04
%\href{/spires/find/hep/www?j=phlta\%2cb91\%2c152}{SPIRES entry}

%\cite{Bartel:1979ut}
\item%{Bartel:1979ut}
{\bf ``Observation Of Planar Three Jet Events In E+ E- Annihilation And Evidence For Gluon Bremsstrahlung''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 91}, 142 (1980)
%\\{}DESY 79/80
%\href{/spires/find/hep/www?j=phlta\%2cb91\%2c142}{SPIRES entry}

%\cite{Bartel:1979pi}
\item%{Bartel:1979pi}
{\bf ``Search For New Flavor Production At Petra''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 89}, 136 (1979)
%\\{}DESY 79/70
%\href{/spires/find/hep/www?j=phlta\%2cb89\%2c136}{SPIRES entry}

%\cite{Bartel:1979zn}
\item%{Bartel:1979zn}
{\bf ``Total Cross-Section For Hadron Production By E+ E- Annihilation At Petra Energies''}
  \\{}W.~Bartel {\it et al.}  [JADE Collaboration]
  \\{}Phys.\ Lett.\  B {\bf 88}, 171 (1979)
%\\{}DESY 79/64
%\href{/spires/find/hep/www?j=phlta\%2cb88\%2c171}{SPIRES entry}


\end{enumerate}
\end{document}

